

Author's response to reviews

Title: Transcription factor E2F1 promotes EMT by regulating ZEB2 in small cell lung cancer

Authors:

Zunling Li (lizunling@bzmc.edu.cn)

Tingting Wang (christinawang0206@163.com)

Xufang Chen (ytchenxf@163.com)

Weiwei Qiao (qiaoweimei@126.com)

Lijun Kong (yt_klijun@163.com)

Daqing Sun (sundaqingchris@126.com)

Version: 1 Date: 27 Jul 2017

Author's response to reviews:

Dear Editor:

We are very grateful to you and the peer reviewers for the valuable comments. According to these comments, we carefully revised our manuscript and added a lot of data to better support our conclusion. We will attach our point-by-point response and our revised manuscripts. We used the red to highlight our changes in the manuscript.

We express our gratitude once again for you and the peer reviewers.

Sincerely,

Zunling Li, Ph D.

Corresponding author

Department of Biochemistry and Molecular Biology

Binzhou Medical University

346 Guanhai Road

Yantai, Shandong 264003

Phone: 86-535-6913211

E-mail: lizunling@bzmc.edu.cn